Title: Periodontal conditions, oral Candida albicans and salivary proteins in type 2 diabetic subjects with emphasis on gender.

Authors:

Fawad Javed (fawad.javed@ki.se)
Lena Klingpor (lena.klingpor@karolinska.se)
Ulf Sundin (ulf.sundin@karolinska.se)
Mohammad Altamash (draltamash@cyber.net.pk)
Björn Klinge (bjorn.klinge@ki.se)
Per-Erik Engström (per-erik.engstrom@ki.se)

Version: 8 Date: 8 May 2009

Author's response to reviews: see over
To,

The Editor,
BMC Oral Health

Subject: Final submission of accepted Manuscript (ID: 3328842932290015)

Dear Sir,

Thank you for sending the pdf version of our accepted manuscript. After a careful review of the pdf version, we have noticed that there are a few alterations that need to be made in the text. These changes are listed as follows:

1. Page 8: Periodontal examination, number of teeth and denture-wearing. Please write the formula for percentage of the numbers of teeth as under:

   \[(\text{Number of present teeth} \div 28) \times 100\]

2. Figure 3: The Y-axis shows the concentration in ug/mg. It is not clearly written on the Y-axis.

3. Figure 4: The bars for type 2 diabetic subjects without oral \textit{C. albicans} do not contain the stripes and dots as shown in the Key.

Thank you.

Yours sincerely,

\[\text{Fawad Javed} \]
\text{Corresponding author}
Karolinska Institutet,
Department of Dental Medicine, Division of Periodontology,
Box 4064, SE 141 04, Huddinge, SWEDEN.
Ph (Off.): +46 8 524 88 225     Fax: +46 8 711 83 43     Email: fawad.javed@ki.se